Author's response to reviews

Title: Prevalence of human papillomavirus in penile malignant tumors: viral genotyping and clinical aspects

Authors:

Isaura Danielli Borges de Sousa (isauradanielli@gmail.com)
Flavia Castello Branco Vidal Dr (flavidal@hotmail.com)
Joao Paulo Castello Branco Vidal (jpvidal@gmail.com)
George Castro Figueira Mello Dr (geocfdmello@gmail.com)
Maria do Desterro Soares Brandao Nascimento Dr (cnsd_ma@uol.com.br)
Luciane Maria Oliveira Brito Dr (luciane2406@yahoo.com.br)

Version: 2
Date: 22 October 2014

Author's response to reviews: see over
Dear Dr. Hayley Henderson,

Please find enclose a new version of our manuscript “Prevalence of human papillomavirus in penile malignant tumors: viral genotyping and clinical aspects”, to be evaluated in BMC Urology. In this new version, we included line and page numbers. We did not include a list of abbreviations because we don’t have any. We stated in the Methods section that as the samples consisted in paraffined tumours, there was no written informed consent from the patients. The patient’s identity was not disclosed in this research (page 4, lines 88-90).

Sincerely yours,

Flávia Vidal, PhD
Professor, Department of Morphology
Federal University of Maranhão